Author’s response to reviews

Title: Genetic Variants in CYP4F2 were Significantly Correlated with Susceptibility to Ischemic Stroke

Authors:
Yuan Wu (yiyi7229@163.com)
Junjie Zhao (zhaojunjie_2009@126.com)
Yonglin Zhao (zhaoyonglinlin@163.com)
Tingqin Huang (guge88@qq.com)
Xudong Ma (747001758@qq.com)
Honggang Pang (646613359@qq.com)
Ming Zhang (zhangming_361@163.com)

Version: 2 Date: 26 Aug 2019

Author’s response to reviews:

Dear Editors,

On behalf of my co-authors, we thank you very much for giving us an opportunity to revise our manuscript, we appreciate you very much for your formatting changes on our manuscript entitled “Genetic Variants in CYP4F2 were Significantly Correlated with Susceptibility to Ischemic Stroke” (ID: MGTC-D-19-00166R3). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and made correction. We hope meet with approval.

We have carefully read the editor’s critical suggestions and revised the manuscript point-by-point. Revised portion are marked using the track changes mode in the marked manuscript.

We appreciate for Editors warm work earnestly. We revised our paper point-by-point and grammatical problems. Finally, we hope that the correction will meet with approval. Looking forward to hearing from you.

Thank you and best regards!

Yours sincerely,
Ming Zhang.